Author's response to reviews

Title: Managing the environmental impact of research

Authors:

   David Pencheon (David.Pencheon@sdu.nhs.uk)

Version: 3 Date: 16 February 2011

Author's response to reviews:

1. Page 2, 3rd line of Introduction. ?reason? should be ?research?.

   Reducing waste of any sort of resource is both an important reason for doing reason and an important part of the process of conducting the research

   ...should read

   Reducing waste of any sort of resource is both an important reason for doing research and an important part of the process of conducting the research

2. page 4, 2nd last line. ?Trail? should be ?Trial?.

   Should read: Trial methodology is an obvious place to start.